Author’s response to reviews

Title: Protocol of a randomized control trial of the Tobacco Tactics Website for operating engineers

Authors:

Sonia A Duffy (bump@umich.edu)
David Ronis (dronis@umich.edu)
Caroline Richardson (caroli@umich.edu)
Andrea H Waltje (waltjea@med.umich.edu)
Lee A Ewing (lee.ewing@va.gov)
Devon Noonan (dnoonan@umich.edu)
Oisaeng Hong (OiSaeng.Hong@nursing.ucsf.edu)
John Meeker (meekerj@umich.edu)

Version: 2 Date: 21 March 2012

Author’s response to reviews:

We have made the following revisions:

Funding: We have stated the funding body (National Cancer Institute) and grant number (1R21CA152247-01A1) of this study on page 5, 2nd paragraph of background section as well as in the background section of the abstract and in the acknowledgments. We have also emailed proof of funding.

Ethical/IRB approval: We have stated that IRB approval was received from the University of Michigan on page 8, design section of the manuscript. We have also emailed proof of funding.

Trial Registration: The trial registration is listed at the conclusion of the abstract.